Author's response to reviews

Title: EuropeaN Energy balance Research to prevent excessive weight Gain among Youth (ENERGY) project: Design and methodology of the ENERGY cross sectional survey.

Authors:

Maartje M van Stralen (mm.vanstralen@vumc.nl)
Saskia J te Velde (s.tevelde@vumc.nl)
Amika S Singh (a.singh@vumc.nl)
Ilse De Bourdeaudhuij (Ilse.Debourdeaudhuij@UGent.be)
Marloes K Martens (m.martens@rescon.nl)
Maria van der Sluis (m.vandersluis@rescon.nl)
Yannis Manios (manios@hua.gr)
Evangelia Grammatikaki (evagram@hua.gr)
Mai JM Chinapaw (m.chinapaw@vumc.nl)
Lea Maes (Lea.Maes@UGent.be)
Elling Bere (elling.bere@hia.no)
Jorgen Jensen (jorgen@foi.dk)
Luis Moreno (lmoreno@unizar.es)
Nata&353;a Jan (drustvo-zasrce@siol.net)
Dénes Molnár (denes.molnar@aok.pte.hu)
Helen Moore (helen.moore@durham.ac.uk)
Johannes Brug (j.brug@vumc.nl)

Version: 5 Date: 24 January 2011

Author's response to reviews: see over
Dear BMC Series Editorial Production Team,

We would like to thank you for your invitation to resubmit our revised manuscript entitled: “EuropeaN Energy balance Research to prevent excessive weight Gain among Youth (ENERGY) project: Design and methodology of the ENERGY cross-sectional survey." for publication in BMC Public Health.

We have as requested checked our manuscript very carefully for typographical changes. In our manuscript retrieved from your website I was however not able to format the tables in such a way that the first row of the table is repeated on the next page (in cases the table was longer than 1 page). In addition, somehow, I was not able to put table 2 right under its title. Hopefully you will be able to do this during the final formatting of the paper. My apologies for this inconvenience.

With kind regards, on behalf of all co-authors

Maartje van Stralen

-------------------------------------
Post doctoral researcher
VU University Medical Center
EMGO Institute for Health and Care Research
Van der Boechorststraat 7
1081 BT Amsterdam
phone: +31 (0)20 444 9684
email: mm.vanstralen@vumc.nl